

City Research Online

City, University of London Institutional Repository

Citation: Hollowell, J., Rowe, R., Townend, J., Knight, M., Li, Y., Linsell, L., Redshaw, M., Brocklehurst, P., Macfarlane, A. J., Marlow, N., et al (2015). The Birthplace in England national prospective cohort study: further analyses to enhance policy and service delivery decision-making for planned place of birth. Health Service and Delivery Research, 3(36), pp. 1-264. doi: 10.3310/hsdr03360

This is the supplemental version of the paper.

This version of the publication may differ from the final published version.

Permanent repository link: https://openaccess.city.ac.uk/id/eprint/12430/

Link to published version: https://doi.org/10.3310/hsdr03360

Copyright: City Research Online aims to make research outputs of City, University of London available to a wider audience. Copyright and Moral Rights remain with the author(s) and/or copyright holders. URLs from City Research Online may be freely distributed and linked to.

Reuse: Copies of full items can be used for personal research or study, educational, or not-for-profit purposes without prior permission or charge. Provided that the authors, title and full bibliographic details are credited, a hyperlink and/or URL is given for the original metadata page and the content is not changed in any way.
 City Research Online:
 http://openaccess.city.ac.uk/
 publications@city.ac.uk

The Birthplace in England national prospective cohort study: further analyses to enhance policy and service delivery decision-making for planned place of birth

Jennifer Hollowell,^{1*} Rachel Rowe,¹ John Townend,¹ Marian Knight,¹ Yangmei Li,¹ Louise Linsell,¹ Maggie Redshaw,¹ Peter Brocklehurst,² Alison Macfarlane,³ Neil Marlow,² Christine McCourt,³ Mary Newburn,⁴ Jane Sandall⁵ and Louise Silverton⁶

 ¹National Perinatal Epidemiology Unit, Nuffield Department of Population Health, University of Oxford, Oxford, UK
 ²Institute for Women's Health, University College London, London, UK
 ³Centre for Maternal and Child Health Research, City University London, London, UK
 ⁴National Childbirth Trust, London, UK
 ⁵Division of Women's Health, King's College London, London, UK
 ⁶Royal College of Midwives, London, UK

*Corresponding author

Declared competing interests of authors: Peter Brocklehurst has received non-financial support and personal fees as a member of Department of Health and Medical Research Council commissioning and funding panels unrelated to this project. Neil Marlow has received personal consultancy fees from Novartis and Shire for work unrelated to this project.

Published August 2015 DOI: 10.3310/hsdr03360

Plain English summary

The Birthplace in England national prospective cohort study Health Services and Delivery Research 2015; Vol. 3: No. 36 DOI: 10.3310/hsdr03360

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

n England, healthy women with straightforward pregnancies can choose to give birth at home, in freestanding midwifery units, in alongside midwifery units and in consultant-led hospital units (obstetric units). Women with certain health problems or who have had problems in a previous birth are usually advised to give birth in an obstetric unit.

The Birthplace study answered questions about the safety of different places to have a baby; in this follow-on project we carried out five more research studies using some of the information collected in the Birthplace study.

Some of these studies provide more information to help healthy women with straightforward pregnancies decide where to give birth. For example, one study shows that, for women having their first baby, being older or more than 1 week past their due date increases their chances of being transferred to hospital during labour or immediately after birth. Another study shows that for healthy women with straightforward pregnancies, whatever their age, ethnic background or the type of area they live in, planning to have their baby outside a hospital obstetric unit means that they are less likely to have medical intervention during birth, such as drugs to speed up labour, or forceps or ventouse to help deliver the baby.

For women with more complicated pregnancies, we looked at what happens to women who plan to give birth at home compared with women who plan to give birth in an obstetric unit. We found some differences between these two groups of women in how often their babies are admitted to special care after birth, but our study was too small to answer questions about uncommon, but serious, poor outcomes for the babies.

We also looked at the way in which maternity services are organised. We found that some aspects of the organisation of services may make small differences to things that happen during labour and birth, including transfer to an obstetric unit and how likely a woman is to have a caesarean section.

Many of the findings from this project and the further research that we recommend are intended to help the NHS plan better services and improve information given to women.

Health Services and Delivery Research

ISSN 2050-4349 (Print)

ISSN 2050-4357 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: nihredit@southampton.ac.uk

The full HS&DR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Services and Delivery Research journal

Reports are published in *Health Services and Delivery Research* (HS&DR) if (1) they have resulted from work for the HS&DR programme or programmes which preceded the HS&DR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

HS&DR programme

The Health Services and Delivery Research (HS&DR) programme, part of the National Institute for Health Research (NIHR), was established to fund a broad range of research. It combines the strengths and contributions of two previous NIHR research programmes: the Health Services Research (HSR) programme and the Service Delivery and Organisation (SDO) programme, which were merged in January 2012.

The HS&DR programme aims to produce rigorous and relevant evidence on the quality, access and organisation of health services including costs and outcomes, as well as research on implementation. The programme will enhance the strategic focus on research that matters to the NHS and is keen to support ambitious evaluative research to improve health services.

For more information about the HS&DR programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hsdr

This report

The research reported in this issue of the journal was funded by the HS&DR programme or one of its preceding programmes as project number 10/1008/43. The contractual start date was in August 2012. The final report began editorial review in August 2014 and was accepted for publication in December 2014. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HS&DR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health.

© Queen's Printer and Controller of HMSO 2015. This work was produced by Hollowell *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

Health Services and Delivery Research Editor-in-Chief

Professor Ray Fitzpatrick Professor of Public Health and Primary Care, University of Oxford, UK

NIHR Journals Library Editor-in-Chief

Professor Tom Walley Director, NIHR Evaluation, Trials and Studies and Director of the HTA Programme, UK

NIHR Journals Library Editors

Professor Ken Stein Chair of HTA Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andree Le May Chair of NIHR Journals Library Editorial Group (EME, HS&DR, PGfAR, PHR journals)

Dr Martin Ashton-Key Consultant in Public Health Medicine/Consultant Advisor, NETSCC, UK

Professor Matthias Beck Chair in Public Sector Management and Subject Leader (Management Group), Queen's University Management School, Queen's University Belfast, UK

Professor Aileen Clarke Professor of Public Health and Health Services Research, Warwick Medical School, University of Warwick, UK

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Peter Davidson Director of NETSCC, HTA, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

Professor Elaine McColl Director, Newcastle Clinical Trials Unit, Institute of Health and Society, Newcastle University, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Health Sciences Research, Faculty of Education, University of Winchester, UK

Professor John Norrie Health Services Research Unit, University of Aberdeen, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Institute of Child Health, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of members of the NIHR Journals Library Board: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: nihredit@southampton.ac.uk